Author’s response to reviews

Title: TNFSF15 promoter polymorphisms increase the susceptibility to small cell lung cancer: a case-control study

Authors:

Hui Gao (1032838941@qq.com)
Zeren Niu (m13933393663@163.com)
Zhi Zhang (zhi1969@163.com)
Hongjiao Wu (whj1838074359@163.com)
Yuning Xie (121259335@qq.com)
Zhenbang Yang (bzy108@126.com)
Ang Li (Woainiang521@126.com)
Zhenxian Jia (15131267020@163.com)
Xuemei Zhang (jyxuemei@gmail.com)

Version: 3 Date: 26 Jan 2019

Author’s response to reviews:

Response to editor

1. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Response: Done as your request.

2. Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed before acceptance for publication (e.g. "Discussione").

Response: Sorry for this. We asked Dr. Baitang Ning to review and edit the manuscript. Dr. Ning is a senior research biologist at The National Center for Toxicological Research (NCTR) in USA. He is responsible for leading a research team focused on molecular pharmacology and toxicology in the application of personalized medicine.
3. Please note, the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript should be declared in the Funding section.

Response: Done as your request.

4. Please represent authors' names using their initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Response: Done as your request.

5. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of all authors do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. Authors are expected to fulfil the criteria below (adapted from McNutt et al., Proceedings of the National Academy of Sciences, Feb 2018, 201715374; DOI: 10.1073/pnas.1715374115; licensed under CC BY 4.0):

Each author is expected to have made substantial contributions to the conception OR design of the work; OR the acquisition, analysis, OR interpretation of data; OR the creation of new software used in the work; OR have drafted the work or substantively revised it

AND to have approved the submitted version (and any substantially modified version that involves the author's contribution to the study);

AND to have agreed both to be personally accountable for the author's own contributions and to ensure that questions related to the accuracy or integrity of any part of the work, even ones in which the author was not personally involved, are appropriately investigated, resolved, and the resolution documented in the literature.

Acquisition of funding, collection of data or general supervision of the research group, alone, does not usually justify authorship.

If these guidelines are not met, we would request the following change of authorship form be filled out and sent to our editorial office - https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5
Anyone who contributed towards the article who does not meet the criteria for authorship can be acknowledged in the ‘Acknowledgements’ section.

Response: Thank you for your suggestion. We have rewritten this part.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Done as your request.